Dear Editors,

I enclose a research article that I would like you to consider for publication in your journal. I believe that the report is of general interest to the psychiatric medical community as it is the first time that structural MR imaging has been shown to be potentially useful in distinguishing between individuals at high familial risk who have transient or partial psychotic symptoms who will and will not develop schizophrenia. The lack of any previous test of this kind has limited the applicability of possible management and as such our test, although preliminary, represents a major step towards an approach to the management of schizophrenia.

Yours sincerely

Dominic Job

Division of Psychiatry,
School of Molecular and Clinical Medicine,
The University of Edinburgh,
The Royal Edinburgh Hospital,
Morningside Park,
Edinburgh,
Scotland,
U.K. EH10 5HF